Author's response to reviews

Title: Caseous calcification of the mitral annulus with mitral regurgitation and impairment of functional class: a case report.

Authors:

Giovanni GM Minardi (giovanni.minardi@libero.it)
Carla CM Manzara (cmanzara@scamilloforlanini.rm.it)
Giovanni GP Pulignano (gipulig@yahoo.it)
Paolo G. PGP Pino (ppino@scamilloforlanini.rm.it)
Herribert HP Pavaci (herrpav@hotmail.com)
Martina MS Sordi (martinasordi@yahoo.it)
Francesco FM Musumeci (fmusumeci@scamilloforlanini.rm.it)

Version: 2 Date: 30 November 2007

Author's response to reviews:

November 30th, 2007

Michael Kidd, MD
Editor-in-Chief of Journal of Medical Case Reports

Dear Dr. Kidd,

We are submitting you the revision of our manuscript: Caseous calcification of the mitral annulus with mitral regurgitation and impairment of functional class: a case report, modifying it as suggested, and addressing the standard of grammar and spelling.

We hope for a positive consideration,

Yours Sincerely,

Giovanni Minardi